

**Author's response to reviews**

**Title:** A process evaluation of systematic risk and needs assessment for caregivers in specialised palliative care

**Authors:**

Kia Toft Thomsen (kia.toft@oncology.au.dk)

Mai-Britt Guldin (m.guldin@ph.au.dk)

Mette Nielsen (mette.k.nielsen@ph.au.dk)

Chaitali Ollars (chailaura@hotmail.com)

Anders Jensen (andjense@rm.dk)

**Version: 1 Date:** 27 Feb 2017

**Author's response to reviews:**

The authors' response letter has been included as a supplementary file